Identification and characterisation of a rare *MTTP* variant underlying hereditary non-alcoholic fatty liver disease.

Jane Grove^{1,2,a} Peggy Lo^{3,4,a} Nick Shrine^{5,a} Julian Barwell⁶ Louise Wain^{5,7} Martin Tobin^{5,7} Andrew Salter⁸ Aditi Borkar⁹ Sara Cuevas-Ocana^{3,4} Neil Bennett⁵ Catherine John⁵ Ioanna Ntalla⁶ Gabriela Jones^{6,b} Christopher Neal¹⁰ Mervyn Thomas¹¹ Helen Kuht¹¹ Pankaj Gupta^{12,13} Vishwaraj Vemala¹⁴ Allister Grant^{14,c} Adeolu Adewoye^{15,d} Kotacherry Shenoy¹⁶ Leena Balakumaran¹⁶ Edward Hollox^{15,e} Nicholas Hannan^{3,4,e} Guruprasad Aithal^{1,2,e*}

¹NIHR Nottingham Biomedical Research Centre, Nottingham University Hospitals NHS Trust & University of Nottingham, Nottingham, NG7 2UH, UK.

²Nottingham Digestive Diseases Centre, School of Medicine, University of Nottingham, Nottingham, NG7 2UH, UK.

³Division of Cancer and Stem Cells, School of Medicine, University of Nottingham, Nottingham, NG7 2RD, UK.

⁴University of Nottingham Biodiscovery Institute, University Park, University of Nottingham, NG7 2RD, UK.

⁵Genetic Epidemiology Group, Department of Population Health Sciences, University of Leicester, Leicester, LE1 7RH, UK.

⁶Clinical Genetics Department, University Hospitals Leicester NHS Trust, Leicester, UK. ⁷NIHR Leicester Respiratory Biomedical Research Centre, Glenfield Hospital, Leicester, UK.

⁸School of Biosciences, University of Nottingham, Sutton Bonington Campus, Loughborough, LE12 5RD, UK.

⁹School of Veterinary Medicine and Science, University of Nottingham, Sutton Bonington Campus, Loughborough, LE12 5RD, UK.

¹⁰Leicester Cancer Research Centre, University of Leicester, Leicester UK

¹¹Ulverscroft Eye Unit, Dept. of Neuroscience, Psychology and Behaviour, University of Leicester, Leicester, LE2 7LX, UK.

¹²Department of Chemical Pathology and Metabolic Diseases, University Hospitals of Leicester NHS Trust, Leicester, UK.

¹³Department of Cardiovascular Sciences, University of Leicester, Leicester, UK.

¹⁴Department of Gastroenterology, University Hospitals of Leicester NHS Trust, Leicester, LE1 5WW, UK.

¹⁵Department of Genetics and Genome Biology, University of Leicester, Leicester, UK. ¹⁶Population Health and Research Institute, Trivandrum, India.

^aThese authors contributed equally ^bCurrent address: Department of Genetics, Nottingham University Hospitals NHS Trust, Nottingham, UK. ^cCurrent address: Treliske Hospital, Truro, Cornwall, TR1 3LJ, UK ^dCurrent address: Department of Biosciences, School of Science and Technology, Nottingham Trent University, Clifton Campus, Nottingham, NG11 8NS ^eSenior Authors

* Corresponding author: Guruprasad P. Aithal (ORCID: 0000-0003-3924-4830) Nottingham Digestive Diseases Centre, Queens Medical Centre Campus, Nottingham NG7 2UH, UK. E-mail guru.aithal@nottingham.ac.uk

Key Words: microsomal triglyceride transfer protein; abetalipoproteinemia; hiPSC

derived hepatocytes; lipoprotein ApoB

Word Count= 5989

Figures = 6

Tables = 2

Conflicts of Interest

Guruprasad Aithal has served as a consultant and an advisory board member for Pfizer Inc, Inventiva Pharma, GlaxoSmithKline and KaNDy Therapeutics; he has been a consultant to Servier, Clinipace, Albireo Pharma, BenevolentAI Bio, DNDi, BerGenBio ASA, Median Technologies, FRACTYL, Amryt Pharma and AstraZeneca; and has given presentations on behalf of Roche Diagnostics and Medscape. Ioanna Ntalla is employed by Gilead Sciences Ltd. (since August 2019). All other authors declare no conflict of interests

Financial Support

This work was supported by the Medical Research Council (MRC) Nottingham Molecular Pathology Node [grant number MR/N005953/1], National Institute of Health Research (NIHR) Nottingham Digestive Diseases Biomedical Research Unit and Nottingham Biomedical Research Centre [BRC-1215-20003]. J Grove and G Aithal are supported by NIHR Nottingham Biomedical Research Centre. K Shenoy and L Balakumaran are

supported by Population Health and Research Institute. All cell modelling was supported by the RoseTrees Trust and the Stoneygate Trust [M546]. L. Wain holds a GSK/British Lung Foundation Chair in Respiratory Research (C17-1). The research was supported by the NIHR Leicester Biomedical Research Centre; C. John held a Medical Research Council Clinical Research Training Fellowship [MR/P00167X/1]. EXCEED is supported by the University of Leicester, the NIHR Leicester Respiratory Biomedical Research Centre, by Wellcome [202849] and by Cohort Access fees from studies funded by the Medical Research Council (MRC), BBRSC, NIHR, the UK Space Agency, and GSK. This work is supported by BREATHE - The Health Data Research Hub for Respiratory Health [UKR_PC_19004] in partnership with SAIL Databank. The exome sequencing was funded by MRC Grant Senior Clinical Fellowship to M. Tobin (G0902313) and we thank the highthroughput genomics group at the Wellcome Trust Centre for Human Genetics (funded by Wellcome Trust Grant 090532/Z/09/Z and MRC Hub Grant G090074791070) for the generation of the sequence data. M. Tobin is supported by a Wellcome Trust Investigator Award [WT202849/Z/16/Z] and holds an NIHR Senior Investigator Award. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Author Contributions:

Conceptualization: Aithal

Study Design: Aithal, Barwell, Hannan, Salter, Wain, Tobin.
Data curation: Grove, Lo, Hannan, Hollox, Shrine, Bennett, John, Ntalla.
Funding: Aithal, Hannan, Hollox, Shenoy, Tobin, Wain.
Formal analysis: Grove, Hannan, Lo, Shrine, Bennett.
Investigation: Grove, Lo, Aithal, Hannan, Barwell, Shrine, Borkar, Cuevas-Ocana, Bennett, Hollox, Grant, Jones, Thomas, Kuht, Adewoye, Gupta, Vemala
Supervision: Aithal, Hannan, Barwell, Salter, Grove, Wain, Tobin.
Resources: Aithal, Grove, Hannan, Salter, Barwell, John, Ntalla, Neal, Shenoy, Balakumaran
Interpretation of data: Aithal, Grove, Hannan, Salter, Tobin, Barwell, Wain, Hollox
Writing- original draft: Grove, Lo, Hannan, Salter, Barwell, Wain, Aithal.

Writing- review & editing: all authors

Abstract

Background and aims: Non-alcoholic fatty liver disease (NAFLD) is a complex trait with an estimated prevalence of 25% globally. We aimed to identify the genetic variant underlying a four-generation family with progressive NAFLD leading to cirrhosis, decompensation and development of hepatocellular carcinoma in the absence of common risk factors such as obesity and type 2 diabetes.

Methods: Exome sequencing and genome comparisons were used to identify the likely causal variant. We extensively characterised the clinical phenotype and post-prandial metabolic responses of family members with the identified novel variant in comparison to healthy non-carriers and wild-type patients with NAFLD. Variant-expressing hepatocyte-like cells (HLCs) were derived from human induced pluripotent stem cells generated from homozygous donor skin fibroblasts and restored to wild-type using CRISPR-Cas9. The phenotype was assessed using imaging, targeted RNA analysis and molecular expression arrays.

Results: We identified a rare causal variant c.1691T>C p.I564T (rs745447480) in *MTTP*, encoding microsomal triglyceride transfer protein (MTP), associated with progressive NAFLD, unrelated to metabolic syndrome and without characteristic features of abetalipoproteinemia. HLCs derived from a homozygote donor had significantly lower MTP activity and lower lipoprotein ApoB secretion compared to wild-type cells, while having similar levels of MTP mRNA and protein. Cytoplasmic triglyceride accumulation in HLCs triggered endoplasmic reticulum stress, secretion of pro-inflammatory mediators and production of reactive oxygen species.

Conclusion: We have identified and characterized a rare causal variant in *MTTP* and homozygosity for *MTTP* p.I564T is associated with progressive NAFLD without any other manifestations of abetalipoproteinemia. Our findings provide insights into mechanisms driving progressive NAFLD.

Lay Summary

- A rare genetic variant in the gene *MTTP* has been identified as responsible for development of severe non-alcoholic fatty liver disease in a four-generation family with no typical disease risk factors.
- A cell line culture created harbouring this variant gene was characterized to understand how this genetic variation leads to a defect in liver cells which results in accumulation of fat and processes which promote disease.
- This is now a useful model for studying the disease pathways and to discover new ways to treat common types of fatty liver disease.

Introduction

Non-alcoholic fatty liver disease (NAFLD) is a complex trait encompassing a spectrum of accumulation of triglyceride-rich lipid droplets within the hepatocytes (steatosis), non-alcoholic steatohepatitis (NASH; having ballooning degeneration and inflammatory cell infiltration), varying degree and pattern of fibrosis leading to cirrhosis and its decompensation, as well as hepatocellular carcinoma (HCC). With rising incidence of obesity and type 2 diabetes, NAFLD is now the most common chronic liver disease with an estimated 25% population prevalence globally.¹

Genome-wide association studies (GWAS) have identified a number of genetic risk variants for NAFLD including *PNPLA3* rs738409 and *TM6SF2* rs58542926, both of which have robust associations with disease phenotypes via functional pathobiological pathways.^{2,3} Accretion of the PNPLA3 variant on lipid droplets sequesters coactivators, resulting in reduced lipolysis and lipophagy, and the TM6SF2 variant impairs VLDL lipidation. Accumulation of triglycerides in both contexts is associated with progressive liver disease.²

Microsomal transfer protein (MTP) as a heterodimer with protein disulphide isomerase (PDI) catalyses lipidation and assembly of ApoB-containing lipoproteins for secretion by hepatocytes and *MTTP* variants have been linked with susceptibility to NAFLD^{4,5}. Rare,

loss-of-function mutations in *MTTP* can result in the recessive disorder abetalipoproteinemia⁶, where MTP deficiency causes defective lipoprotein biosynthesis having multiple severe effects including liver steatosis and fibrosis.^{5,7} However, hereditary progressive NAFLD associated with a *MTTP* variant, without any manifestations of abetalipoproteinemia, has not been previously described.

Here we have clinically characterised a large four-generation family found to have a rare *MTTP* variant located at the interface with PDI resulting in progressive NAFLD, with consequent cirrhosis, liver failure and hepatocellular carcinoma in homozygotes. We evaluated post-prandial metabolic responses in carriers of the novel *MTTP* p.I564T variant compared to non-carriers. We used hepatocyte-like cells (HLCs) derived from human induced pluripotent stem cells (hiPSCs) generated from donor skin fibroblasts from carriers and non-carriers of the *MTTP* variant, as a stable reproducible model for understanding the effect of the variant on the cellular phenotype. This has enabled us to understand how disrupted hepatic lipid homeostasis can drive steatosis and NAFLD and therefore linking genotype to phenotype in hereditary NAFLD.

Patients and methods

Further details of methods used are available in Supplementary material.

Human Samples

The clinical studies were approved by the Health Research Authority after review by the National Research Ethics Service: East Midlands Northampton Committee for the Genetics of Rare Inherited Disorders (GRID) study (Ref 12/EM/0262); North-East Committee for meal-response study (Ref: 16/NE/0251). Studies were conducted according to the Declaration of Helsinki (Hong Kong Amendment) and Good Clinical Practice (European guidelines). All participants provided written informed consent. For the meal-response analysis, participants were recruited to the study at Queens Medical Centre, Nottingham University Hospitals between 1/11/2016 and 1/6/2017. Patients with biopsy-proven NAFLD, sex and age matched (within 10 y) to family members, were

consecutively identified from a large secondary care cohort who had previously participated in research, and invited to participate. Healthy volunteers were similarly identified and invited. None had diabetes or hazardous alcohol intake and had no known liver disease and had circulating caspase-cleaved CK18 level below 99 U/L.

Clinical investigations followed standard clinical care and included 6-month follow-up as required. Variants segregating with disease were identified following exome sequencing (llumina HiSeq2000). Genotype determination was done using Sanger sequencing (Source Bioscience Ltd) or PCR restriction fragment analysis.

In silico analysis

Models of MTP were based on Protein Data Bank sequence 617S⁸ and visualised using Visual Molecular Dynamics software.⁹

Metabolite and protein analyses

Serum cholesterol, triglycerides, and ApoB, and plasma glucose were quantified using calibrated Horiba auto-analyser and reagents following validated standard manufacturer protocols (Horiba ABX) at the University of Nottingham Metabolic Analysis Facility. Serum insulin was quantified using Human Insulin specific RAI kit (Millipore). Plasma lipoproteins were separated by sequential non-equilibrium density-gradient ultracentrifugation.

Apolipoprotein B-100 (apoB-100) was determined in culture supernatants by ELISA (Sigma-Aldrich) in duplicate (twice). MTP activity was determined in lysed cells (in triplicate) at 4 dilutions using MTP Activity Assay Kit (Sigma-Aldrich). Human NFkB Pathway, Phospho-Kinase and XL Cytokine Array Kits (R&D Systems) were used to determine protein expression or secretion.

Fibroblast reprogramming, hiPSC maintenance and differentiation

Two 2mm skin punch biopsies were obtained from study participants and primary dermal fibroblasts established and skin fibroblasts were reprogrammed using CytoTune iPS 2.0 Sendai Reprogramming Kit (Invitrogen, ThermoFisher) in accordance with the

manufacturer's guidelines. Mesoderm, ectoderm and hepatocyte differentiation of hiPSCs was as described previously.¹⁰⁻¹²

CRISPR-Cas9 mediated correction of I564T mutation in the MTTP(VAR/VAR)

For CRISPR-Cas9 editing, single-guide RNA (gaacatcctgctgtctactg) was cloned and nucleofected (Lonza) into the *MTTP*^(VAR/VAR) parental line.¹³ Clones were screened to select one with corrected alleles. hiPSCs derived from clones with corrected allele *MTTP*^(WT*/WT*), and the parental line were differentiated to HLCs in parallel for characterisation.

Imaging and analysis of mitochondrial function and cellular reactive oxygen species

Cells were stained using Nile Red, Hoechst, DAPI or antibodies. Mitochondrial content of HLCs were visualised using 100nM MitoTracker green FM or MitoTracker deep red FM and intracellular reactive oxygen species (ROS) and mitochondrial superoxide production were assessed using 2.5µM CellROX green or 2.5µM MitoSox Red. Mitochondrial respiration was determined using the Seahorse XF96 analyser (Seahorse Bioscience, USA).

Gene expression analysis and RNA sequencing

Quantitative real-time PCR (qPCR) was carried out as described in supplemental methods. Fold changes in expression were calculated using comparative $\Delta\Delta$ Ct method standardised against the housekeeping gene PBGD and mean of Ct values±standard error of mean (SEM) reported.¹² RNA sequencing and bioinformatics analysis were performed at the Babraham Institute (Cambridge, UK).

Statistical analysis

Statistical analyses were performed using Graph Pad Prism version 8 (La Jolla, USA) software. One-way ANOVA followed by Dunnett's multiple comparison test were used to compare data from samples grouped by a single factor. Student's T-test was used to compare the means of variables determined in two groups.

Results

Clinical presentation of family

A British four-generation family with recent Indian ancestry (Fig. 1A) was referred for genetic counselling after three individuals from the same generation developed HCC. Parent A had no history of NAFLD or metabolic syndrome; Parent B presented with NAFLD symptoms aged 80 and was diagnosed with cirrhosis aged 87. Clinical investigations found all ten children had NAFLD diagnosis as adults (C-L in Table 1) with progression to NASH, cirrhosis (in seven) and HCC (in four) suggestive of a high conversion rate between NAFLD to cirrhosis and hepatocellular carcinoma. Only one of the affected individuals had BMI>30 and instances of type 2 diabetes, hypertension or hyperlipidaemia within the family were not linked with the presence or severity of disease.

Identification of rare MTTP variant allele associated with diagnosis

Functional variants which were unique to affected family members were identified by whole exome sequencing of twelve affected individuals (Fig. 1A) by comparison with nine unaffected South Asian controls (including spouses of F, G, H & J , three unrelated participants from the EXCEED study¹⁴, and two unrelated South Asian individuals with cholangiocarcinoma) and databases of genetic variation identified in the general population. We identified a missense variant: genomic NC_000004.12:g.99608899T>C, NM_000253.2:c.1691T>C, protein NP_000244.2:p.Ile564Thr, in *MTTP* that was unique to affected family members (Figs. 1B and 1C) and fully segregated with disease phenotype in those individuals analysed. All six homozygous individuals developed cirrhosis and three also developed HCC, while some heterozygotes had no diagnosed disease (Table 1). Presence of both heterozygotes and homozygotes for the rare allele in 3rd generation of the family implies that individual A must also have carried the variant allele. The presence of fatty liver in wild-type individual I is suggested to be incidental relating to lifestyle factors.

This I564T variant has been previously described in combination with a second rare variant (IVS1+1G>C), manifesting as severe fatty liver in an atypical case of abetalipoproteinemia in Japan¹⁵ but was reported to have 'mild effect' in the mother carrying I564T alone. The I564T variant is described in the NCBI database¹⁶ as rs745447480 with allele frequency <1.6 x 10⁵ in an analysis of 251,024 alleles (gnomAD exomes v2.1.1) present in four non-Finnish European cases. The NCBI allele frequency aggregator population database reports 2 variants out of a 35,910 global total (both were in European population). GEM-Japan whole genome aggregation panel reports 1 allele in 15,198¹⁷. Other family members were subsequently tested for this variant and clinical features of abetalipoproteinemia⁶ were investigated (Tables 1 & 2). Their genotype for other common functional variants at loci in *MTTP* (rs745447480, rs3816873, rs2306985), *PNPLA3* (rs738409) and *TM6SF2* (rs58542926) associated with NAFLD was also determined. None of 83 NAFLD patients from the Trivandrum cohort¹⁸ had the *MTTP* p.I564T variant allele.

Mapping of the I564T variant onto the crystal structure⁸ located it to heterodimer interaction interface within a hydrophobic pocket on the MTP subunit surface having surrounding polar and charged residues. (Fig. 1C & D). Mutation analysis predicts medium mutation sensitivity and substitution to threonine, a polar residue, will likely destabilise this hydrophobic pocket, promote interactions with the surrounding polar/charged residues and thereby cause local conformational variations (PolyPhen=1; GERP=5.120; CADD=20.4, Mutation Assessor=0.76, REVEL=0.519)¹⁹. A plausible impact on dimer formation and MTP functionality is therefore expected. In contrast, substitutions E98D, N166S, I128T and H297Q arising from common SNPs, map to the protein surface exposed to the solvent in the complex structure (Fig. 1D) and mutation analysis predicts low mutation sensitivity and high tolerance of substitutions at each of these positions.

Postprandial responses in affected individuals

The phenotypic impact of the *MTTP* p.I564T variant was assessed through investigation of metabolic responses to fat consumption. Family members were invited and responses

in five participants were compared to that in age and gender matched healthy volunteers and NAFLD patients (Fig. 2A; Supplementary Table 1). The level of ApoB, the protein constituent assembled into chylomicron and very low density lipoproteins (VLDL) via the activity of MTP, was notably lower in the MTP564-TT homozygote F than in other participants including two MTP564-IT heterozygotes (participants K and Q) and the MTP564-TT liver-transplant recipient J (Fig. 2B). Subsequent testing of six further heterozygotes also showed levels within the normal range. Testing of a stored, pretransplant serum sample from individual J and clinical data revealed that levels were also markedly lower before receiving a replacement liver where the gene is likely restored.

We also compared levels of serum ApoB-100, the isoform associated with VLDL, in the MTP564-TT homozygotes F and J, with that of wild-type MTP564-II control, participant 1, supporting the proposal that expression of the variant form in the liver results in reduced VLDL secretion (Fig. 2C). Levels of total cholesterol both before and after the meal were also noticeably lower in the MTP564-TT homozygote F (Supplementary Fig. 1) compared to all other study participants due to only very low levels of HDL-cholesterol being present (<0.4mmol/L). Clinical data confirms this observation (Table 1) and the same phenotype was noted in another MTP564-TT homozygote, H; however levels in homozygote J, pre-transplant, were normal. Participant F also reported post-prandial gastrointestinal discomfort and diarrhoea following the study meal.

The levels of circulating triglyceride in F were lower than in the healthy and disease controls (participants 1 and 2, Fig. 3A) but were similar to levels in participant 3 who possessed two variant alleles for *TM6SF2* rs58542926 (TM6SF2-KK). TM6SF2 is also involved in VLDL lipoprotein secretion and the variant associated with impaired function and decreased serum LDL-cholesterol.^{20,21} In contrast, the MTP564-TT homozygote with a liver transplant showed a similar triglyceride response to her two matched controls (Fig. 3B). Investigation of VLDL and chylomicron lipoprotein associated triglycerides (Fig. 3C-F) revealed that both components were again lowered in participant F (MTP564-TT) and 3 (TM6SF2-KK) compared to the controls while appeared normal the transplanted MTP564-TT, J. Other family members had no notable defects in lipoprotein triglycerides

(Supplementary Fig. 2). Furthermore, VLDL-cholesterol levels were similarly blunted in participants F (MTP564-TT) and 3 (TM6SF2-KK) but not in other family members (Supplementary Fig. 3) suggesting that transplant hepatocytes and MTP564-IT heterozygote hepatocytes are functioning effectively in VLDL secretion.

Of note, lipoprotein-associated lipid levels are also lower in participant 7, a NAFLD patient who is homozygous for the *PNPLA3* rs738409 variant (PNPLA3-MM) which has been linked to a relative reduction in large VLDL secretion²² (Supplementary Figs. 2A and 3F). Circulating free fatty acids, glucose and insulin levels in the family members were unremarkable (Supplementary Fig. 4).

Generation of wild-type and MTTP^(VAR/VAR) hiPSCs for disease modelling

To elucidate the mechanisms driving hepatic steatosis in homozygous MTP564-TT patients, we generated hiPSCs and differentiated them into hepatocytes to create an in vitro model of the variant. Fibroblasts derived from participant 1, genotyped as MTP564-II (also PNPLA3-148-IM; TM6SF2-167-EE), and family member J, MTP564-TT (also PNPLA3-148-IM; TM6SF2-167-EE), were expanded up to passage 4 and then reprogrammed into hiPSCs: *MTTP*^(WT/WT) and *MTTP*^(VAR/VAR), respectively (Supplementary Figs. 5-7). Neither carry the TM6SF2 variant that could confound the observed phenotype. Reprogrammed fibroblasts displayed the typical features of hiPSCs forming dense cell colonies, with well-defined colony boarders, containing cells with a high nuclear to cytoplasm ratio. To confirm their pluripotent status, Oct3/4 and NANOG expression was assessed and their ability to differentiate into endoderm, mesoderm and ectoderm determined. We confirmed the karyotype as normal, without any major chromosomal abnormalities.

MTP levels are lower and lipoprotein secretion is impaired in *MTTP*^(VAR/VAR) HLCs compared to *MTTP*^(WT/WT) HLCs

To prevent bias due to differing differentiation efficiency of hiPSC lines, we differentiated cells from the study donors 1 and J into HLCs to compare morphology and gene expression profiles.¹² Both cell lines appeared morphologically similar during all stages of differentiation and generated a monolayer of HLCs by day 21 (Fig. 4A). Gene expression

was similar at each of the developmental time points including definitive endoderm, foregut endoderm, and hepatoblast cells. Expression of genes associated with a mature hepatocyte phenotype was not significantly different between the two cell lines (Supplementary Fig. 7). Analysis of mRNA expression patterns for both cell lines primarily matched 'liver bulk tissue' then 'hepatocyte' (Supplementary Table 2) and both showed high similarity to HepG2, HUH7 and HEP3B cell lines.

Nile Red and Oil-Red-O staining revealed phenotypical differences with apparent significant sequestration of lipid via development of discrete lipid droplets and microvesicular steatosis throughout the cytoplasm in the *MTTP*^(VAR/VAR) HLCs after 48h of culture (Fig. 4B,C). Quantification showed levels were >4-fold higher in *MTTP*^(VAR/VAR) HLCs, (Fig. 4E), consistent with the proposed reduced VLDL secretion in cells expressing MTP564-TT, restricting removal of intracellular triglycerides.

Importantly however, although both immunocytochemistry and mRNA expression analysis suggested a trend towards lower MTP levels compared to cells expressing the wild-type allele this was not statistically significant (Fig. 4D,F,G). To assess VLDL export capabilities of the cell lines, and thus functioning of variant MTP in lipoprotein biosynthesis, levels of secreted ApoB-100 were determined. There was significantly less, but detectable, ApoB-100 in the media from *MTTP*^(VAR/VAR) compared to *MTTP*^(WT/WT) HLCs (Fig. 4H) confirming the clinical phenotype and supporting the suggestion that the MTP variant in these patients affects lipid trafficking.

Increased generation of ROS and altered mitochondrial respiration in *MTTP*^(VAR/VAR) HLCs

Hepatic free fatty acids can be converted to triglyceride for storage as cytoplasmic droplets or secreted as VLDL, or else directly metabolised via mitochondrial β-oxidation. Therefore, impaired MTP functionality restricting lipid secretion, thus increasing the availability of fatty acids, may impact on mitochondrial activities. Using mitochondrial stress testing measuring the oxygen consumption rate in live cells revealed *MTTP*^(VAR/VAR) HLCs had significantly higher basal and maximal mitochondrial respiration compared to mitochondria from the wild-type cell line (Figure 4I). Of importance, increased β-

oxidation would generate additional ROS, which can be a major driver of oxidative stress and cellular dysfunction. Both mitochondrial superoxide production and cytoplasmic ROS were significantly higher in the *MTTP*^(VAR/VAR) HLCs compared to *MTTP*^(WT/WT) (Fig. 4J,K and Supplementary Fig. 8) consistent with increased fatty acid metabolism.

Increased NFκB signalling, inflammation, ER stress and secretion of proinflammatory mediators in *MTTP*^(VAR/VAR) HLCs

Impaired lipid trafficking and lipoprotein assembly incurred as a consequence of reduced MTP functionality is likely to cause a range of cellular responses including ER stress and inflammation. Analysis of mRNA revealed that expression of ER stress mediators: spliced X-Box Binding protein-1 (SxBP1); activating transcription factor 6 (ATF6); binding immunoglobulin protein (BIP); and the ER stress transducer, inositol requiring enzyme 1 (IRE1) were significantly higher in *MTTP*^(VAR/VAR) HLCs (Fig. 5A).

MTTP^(VAR/VAR) HLCs expressed significantly higher levels of NFkB (Fig. 5B), suggesting greater ER stress and active pro-inflammatory response.

These observations were confirmed by assessing changes in intracellular and extracellular signalling (Fig. 5C-E). The observed lipid accumulation in the novel MTP564-TT variant strain was associated with increases in NFkB pathway components, indicative of activation, and phosphorylation of pro-inflammatory and pro-apoptotic pathway mediators including RelA/p65 complex, ASC, p53, FADD, and CD95. This increased pro-inflammatory signalling coincided with increased secretion of pro-inflammatory mediators including CCL3/4, BAFF, CD30, IL-1, IL-10 and IL- 23, DBP and leptin while showing decreased RBP4 and PDGF. Notably, *MTTP*^(VAR/VAR) HLCs had more than 10-fold lower expression of phosphorylated-eNOS compared to *MTTP*^(WT/WT) HLCs. There were no significant differences in protein phosphorylation in the other 44 proteins assayed. However, there was a trend towards decreased ERK1/2 phosphorylation in *MTTP*^(VAR/VAR) HLCs.

MTTP^(VAR/VAR) HLCs show increased expression of ECM remodelling and lipid metabolising genes.

Bioinformatics analysis of mRNA sequencing data to assess genome wide changes in gene expression revealed 472 genes differentially expressed (>1x log2-fold) between *MTTP*^(VAR/VAR) and *MTTP*^(WT/WT) cultured HLCs. These fulfil diverse cellular functions including glycolysis, lipid oxidation, oxidative phosphorylation and complement activation (Supplementary Fig. 9) and support previous observations of increased ROS generation and altered mitochondrial activity. Gene Ontology (GO) terms associated with the changing genes (Supplemental Tables 3-5) were mostly implicated in extracellular matrix (ECM) remodelling, ECM organisation and degradation, ECM receptor interactions, and proteoglycan modifications, suggesting that ECM remodelling may be initiated during hepatosteatosis.

Confirmation that the homozygous rs745447480 variant in the *MTTP*^(VAR/VAR) HLCs results in significantly lower MTP lipid transfer activity.

A third cell line, *MTTP*^(WT*/WT*), was generated from *MTTP*^(VAR/VAR) in which the MTP564-TT in was gene-edited to wild-type 564-II using CRISPR-Cas9 transfection and selection of a corrected cloned (Supplemental Fig. 10). The resultant differentiated cell line displayed the same characteristics as the *MTTP*^(WT/WT) cell line (Fig. 6 and Supplemental Fig. 11). This enables us to rule out possibility of other genetic variants harboured by the patient or healthy volunteer influencing the observed in vitro phenotype. MTP lipid transfer activity of the original patient-derived HLCs, *MTTP*^(VAR/VAR), was compared to that of the gene-edited HLCs to establish the impact of the SNP rs745447480. For equivalent cellular protein quantity, MTP activity was significantly lower in the *MTTP*^(VAR/VAR) HLCs compared to the edited derivative *MTTP*^(WT*/WT*) having only 61% of the level determined in the wild-type cells. This is distinct from other described variants which abolish MTP activity and may explain the observed apparent phenotype of impaired lipid trafficking. To assure that the observed phenotype does not reflect compensatory activity of TM6SF2 in the same pathway, TM6SF2 mRNA expression levels were determined. This indicated that TM6SF2 expression, normalised to undifferentiated cells, is not higher in

MTTP^(VAR/VAR) compared to *MTTP*^(WT*/WT*) and therefore suggests that the observed phenotype is not due to increased TM6SF2 (Supplemental Fig. 12).

Discussion

We have identified and characterised a rare MTTP variant (p.I564T) as causative for the Mendelian trait associated with an inherited form of NAFLD, in a four-generation family. Our investigation has revealed a variant resulting in decreased ApoB-containing lipoprotein secretion in homozygotes (but not heterozygotes), in contrast to other variants causing abetalipoproteinemia where ApoB is undetectable (Fig. 2, Table 1)^{6,7,23-} ²⁵. Although other carriage of this variant have been reported, no phenotypic characteristics related to these are previously described¹⁵, none of the GWAS so far including that using UK biobank²⁶ and largest cross-ancestry GWAS²⁷ have identified particular *MTTP* variant (p.I564T) in association with NAFLD. Protein modelling (Fig. 1C) suggests the substitution moderately affects the protein structure and likely impacts upon the interaction with PDI in the formation of a normal heterodimeric enzyme, but is unlikely to abolish all functionality as in abetalipoproteinemia. Presentation of homozygote cases is clearly distinct from abetalipoproteinemia ^{23,28} supporting suggested subtle phenotype whereby impact is limited to liver lipid imbalance. This provides potential for treatment through reduced dietary fat intake and makes it an attractive model for cellular consequences of lipid accumulation.

Our phenotyping studies demonstrated distinct VLDL secretion responses following meal challenge and ApoB levels in the two MTTP-564TT family members: while these biomarkers were substantially low in untreated individual F, these were in the normal range in the liver transplant recipient J (Figure 2b; Table 1). Previously, the *MTTP* -493 variant (rs1800591) G allele linked with reduced MTP function has been associated with NAFLD susceptibility in a meta-analysis of 11 case-control studies.²⁹ Also an association study in non-diabetic patients with NASH found GG homozygotes had significantly higher plasma triglycerides, intestinal and hepatic large VLDL and oxidised LDL than GG/GT group.³⁰ All of the five MTP564-IT heterozygous individuals showed ApoB and lipoprotein

levels within the normal range consistent with reports that a single copy of MTTP is sufficient.³¹

Functional analysis of a childhood case with compound heterozygosity for MTTP c.619-5_619-2del and p.L435H, having severe liver fibrosis but no typical abetaliporoteinemia symptoms, found that p.L435H abolished MTP activity while the intronic variant resulted in 26% of transcripts being normally spliced allowing limited MTTP expression.³² This suggested that the residual expression and resulting MTP activity was sufficient for substantial biological activity covering majority of necessary functionality, except for liver functions. Consistent with this, our observed modest reduction in activity to 61% would thus be predicted to have a subtle effect only on liver homeostasis. The modest impact on protein function in vitro is compatible with the less severe, nonabetalipoproteinemia clinical phenotypes described. The observed `normal' levels of expression in HLCs may suggest that the I564M change effects either translation, protein stability/turnover or enzymatic function. Our in silico models suggested that the impact is on heterodimer assembly, stability, interaction, or activity. The latter would be entirely compatible with the unchanged protein levels observed in HLCs.

Disruption of ApoB biosynthesis and associated VLDL secretion has been widely described with a spectrum of consequences linked to characterised pathologies.³³ ApoB missense variants are also associated with development of fibrosis and HCC linked to NAFLD³⁴. Furthermore, rare variants in MTTP were found to be associated with increased hepatic fat in the UK Biobank cohort⁵. The underlying mechanisms are inherently linked to nutritional intake with diets high in fats (increasing hepatic fat content) and carbohydrate (increasing hepatic de novo lipogenesis) resulting in hyperlipidemia. Hepatic lipid balance is dependent on secretion of VLDL, which is limited by availability/activity of MTP, so any variants with altered activities are likely to have metabolic effects. The low frequency of the MTTP p.I564T variant reported in large datasets¹⁶ means that identification, recruitment and analysis of further carriers to strengthen the study would be very difficult.

Overall, VLDL secretion may increase with hepatic steatosis related to metabolic syndrome.³⁵ However, decreased VLDL secretion has been reported in carriers of PNPLA3 G allele²². VLDL secretion is also lowered in *TM6SF2* T carriers^{20,21,36} which affects the same pathway. We considered key genetic risk factors, PNPLA3 and TM6SF2 variants, likely to influence disease phenotype since polygenic scores have been proposed for NAFLD³⁷. TM6SF2 p.E167K is of particular interest since it acts in the same pathway as MTP so similarities in phenotypes would be expected. In our study, postprandial secretion of VLDL, the predominant postprandial lipoprotein associated with hyperlipidemia³⁸, was lower in participants homozygous for TM6SF2 or PNPLA3, but fasting ApoB levels were normal. Whereas in MTTP p.I564T homozygotes we observed both a reduced level of circulating ApoB and VLDL-associated lipids. We have specifically considered the possibility of functional redundancy between TM6SF2 and MTP and potential additive effects of the variants in vivo and in vitro. First, the clinical characterisation in Table 1 shows the disease phenotype is not linked to TM6SF2 carriage as 3 affected siblings do not carry the TM6SF2 variant. Second, the post-prandial lipid data is consistent with reduction in circulating lipids due to lack of either wild type MTTP or lack of wild type TM6SF2, whereas heterozygotes retain functionality.

In addition to demonstrating the functional consequences of the *MTTP* p.1564T variant, the HLCs derived from hiPSCs provide a disease model for early stage NAFLD, beyond triglyceride accumulation. As the donors are wild type for TM6SF2, the cell phenotype reflects only the impact of the MTTP variant. Studies have shown a link between the amount of steatosis, fibrosis development and liver disease mortality³⁹ with lipid metabolism acting as the initiator of progression to NASH.⁴⁰ Although triglyceride sequestration may be protective, but when fatty acid storage and disposal routes reach capacity, alternative pathways resulting in lipotoxicity can occur. Components of these pathways, such as ACC-1/2, FXR/FGF19/FXR4, and SCD-1, are thus being tested as therapeutic targets.⁴¹ Increased mitochondrial fatty acid β-oxidation may provide a protective response but uncontrolled results in the generation of ROS which can be a major driver of oxidative stress and cellular dysfunction (Figure 4I, Figure 6).

We show that as lipid accumulation increases, hepatocytes have increased ER stress, activate pro-inflammatory signalling pathways including NFκB, P53, eNOS and secrete pro-inflammatory mediators. This coincides with increased production of reactive oxygen species, superoxide production and alterations to mitochondrial respiration driving the disease progression leading to cirrhosis and hepatocellular carcinoma as seen among the family members. Similar findings were reported in cardiomyocytes derived in an MTTP pR46G variant model.⁴² Excessive lipid accumulation in hepatocytes can serve as substrates for the generation of lipotoxic species. One of the major consequences of hepatic lipid metabolism is mitochondrial β-oxidation and esterification to form triglycerides which can serve as a protective mechanism against lipotoxicity in hepatocytes. However, if lipid accumulation is in excess of the β-oxidation capacity, such as in NAFLD, toxic intermediates can accumulate which induce metabolic stress and subsequent inflammation and cell death. Changes in expression of ECM remodelling-associated genes, suggestive of its occurrence during steatosis may contribute to drive progression to fibrosis, which is clinically observed later.

We conclude that the main feature of the *MTTP* p.I564T variant is impaired ApoB secretion and hepatic lipid accumulation due to decreased lipid transfer activity distinct from classical abetalipoproteinemia phenotype where MTP expression is abolished. Identification and characterization of a rare disease such as hereditary NAFLD is of medical significance in Indian populations where high rates of founder events have been reported⁴³. In addition, HLC modelling supports this, providing additional details of signalling, inflammatory and metabolic cellular pathways involved, highlighting pathophysiology driving NAFLD progression and possible therapeutic targets.

Data Availability

Study data are available on request. The three EXCEED exome sequences are available in the European Genome-phenome Archive using accession number EGAD00001007649. Access to sensitive genetic data and cell lines will be restricted to research facilities with institutional data and material transfer agreements to protect participant anonymity.

Abbreviations

ECM: extracellular matrix ELISA: enzyme-linked immunosorbent assay GO: Gene Ontology GWAS Genome-wide association studies HCC: hepatocellular carcinoma hiPSCs: human induced pluripotent stem cells HLCs: hepatocyte-like cells MTP: microsomal triglyceride transfer protein NAFLD: Non-alcoholic fatty liver disease PCR: polymerase chain reaction PDI: protein disulphide isomerase ROS: reactive oxygen species SEM: standard error of mean VLDL: very low density lipoproteins

Acknowledgements

The views expressed are those of the authors and not necessarily those of the National Health Service (NHS), the NIHR or the Department of Health. We thank all the research participants particularly the family involved. We are grateful to the study teams of the EXCEED study, the Trivandrum cohort and NASH study for their contributions. We are grateful to the clinical team at University Hospitals Leicester NHS Trust for clinical work-up and acknowledge support from the late Roger Williams. We thank the high-throughput genomics group at the Wellcome Trust Centre for Human Genetics for the generation of the sequence data. We thank Sally Cordon and Ian Macdonald for assistance with metabolic analysis and Melanie Lingaya and Calum Greenhalgh for technical support. We thank Ester Burden-Teh and Jane Chalmers for assistance with taking skin biopsies and Antonella Ghezzi for obtaining clinical samples for genotyping. We thank Beth Robinson and Nottingham Digestive Diseases Team for assistance with coordinating participant involvement and the meal study. We also thank all participants and staff who have contributed their time to the study.

References

- 1. Younossi ZM, Koenig AB, Abdelatif D, Fazel Y, Henry L, Wymer M. Global epidemiology of nonalcoholic fatty liver disease-Meta-analytic assessment of prevalence, incidence, and outcomes. *Hepatology*. 2016;64(1):73-84.
- Krawczyk M, Liebe R, Lammert F. Toward Genetic Prediction of Nonalcoholic Fatty Liver Disease Trajectories: PNPLA3 and Beyond. *Gastroenterology*. 2020;158(7):1865-1880.e1861.
- 3. Trépo E, Valenti L. Update on NAFLD genetics: From new variants to the clinic. *Journal of hepatology.* 2020;72(6):1196-1209.
- 4. **Tan J,** Zhang J, **Zhao Z**, et al. The association between SNPs rs1800591 and rs3816873 of the MTTP gene and nonalcoholic fatty liver disease: A meta-analysis. *Saudi journal of gastroenterology : official journal of the Saudi Gastroenterology Association.* 2020;26(4):171-178.
- 5. **Haas ME, Pirruccello JP, Friedman SN**, et al. Machine learning enables new insights into genetic contributions to liver fat accumulation. *Cell Genomics*. 2021;1(3):100066.
- 6. Shoulders CC, Brett DJ, Bayliss JD, et al. Abetalipoproteinemia is caused by defects of the gene encoding the 97 kDa subunit of a microsomal triglyceride transfer protein. *Human molecular genetics.* 1993;2(12):2109-2116.
- 7. Di Filippo M, Moulin P, Roy P, et al. Homozygous MTTP and APOB mutations may lead to hepatic steatosis and fibrosis despite metabolic differences in congenital hypocholesterolemia. *Journal of hepatology.* 2014;61(4):891-902.

- 8. Biterova EI, Isupov MN, Keegan RM, et al. The crystal structure of human microsomal triglyceride transfer protein. *Proceedings of the National Academy of Sciences.* 2019;116(35):17251.
- 9. Humphrey W, Dalke A, Schulten K. VMD: visual molecular dynamics. *Journal of molecular graphics.* 1996;14(1):33-38, 27-38.
- 10. **Bertero A, Pawlowski M**, Ortmann D, et al. Optimized inducible shRNA and CRISPR/Cas9 platforms for in vitro studies of human development using hPSCs. *Development.* 2016;143(23):4405.
- 11. Vallier L, Touboul T, Chng Z, et al. Early cell fate decisions of human embryonic stem cells and mouse epiblast stem cells are controlled by the same signalling pathways. *PloS one.* 2009;4(6):e6082.
- 12. Hannan NR, Segeritz CP, Touboul T, Vallier L. Production of hepatocyte-like cells from human pluripotent stem cells. *Nature protocols.* 2013;8(2):430-437.
- 13. Ran FA, Hsu PD, Wright J, Agarwala V, Scott DA, Zhang F. Genome engineering using the CRISPR-Cas9 system. *Nature protocols.* 2013;8(11):2281-2308.
- John C, Reeve NF, Free RC, et al. Cohort Profile: Extended Cohort for E-health, Environment and DNA (EXCEED). *International Journal of Epidemiology*. 2019;48(3):678-679j.
- 15. Sakamoto O, Abukawa D, Takeyama J, et al. An atypical case of abetalipoproteinaemia with severe fatty liver in the absence of steatorrhoea or acanthocytosis. *Eur J Pediatr.* 2006;165(1):68-70.
- 16. Nucleotide [Internet]. Bethesda (MD): National Library of Medicine (US) NCfBIANN, Homo sapiens chromosome 4; [cited 2020 Jun 26]. Available from: <u>https://www.ncbi.nlm.nih.gov/nuccore/NC_000004.12</u>.
- 17. <u>https://grch38.togovar.org/doc/datasets/gem_j_wga</u> JGMaJPG-JAf. GEM Japan Whole Genome Aggregation (GEM-J WGA) Panel
- 18. Chalmers J, Ban L, Leena KB, et al. Cohort profile: the Trivandrum non-alcoholic fatty liver disease (NAFLD) cohort. *BMJ open.* 2019;9(5):e027244.
- 19. <u>https://www.ensembl.org/Homo_sapiens/Variation/Explore?r=4:99608399-99609399;v=rs745447480;vdb=variation;vf=233514873</u>. accessed 26/6/20.
- 20. Prill S, Caddeo A, Baselli G, et al. The TM6SF2 E167K genetic variant induces lipid biosynthesis and reduces apolipoprotein B secretion in human hepatic 3D spheroids. *Scientific reports.* 2019;9(1):11585.
- 21. Kim DS, Jackson AU, Li YK, et al. Novel association of TM6SF2 rs58542926 genotype with increased serum tyrosine levels and decreased apoB-100 particles in Finns. *Journal of lipid research*. 2017;58(7):1471-1481.
- 22. **Pirazzi C, Adiels M**, Burza MA, et al. Patatin-like phospholipase domain-containing 3 (PNPLA3) I148M (rs738409) affects hepatic VLDL secretion in humans and in vitro. *Journal of hepatology.* 2012;57(6):1276-1282.
- 23. Zamel R, Khan R, Pollex RL, Hegele RA. Abetalipoproteinemia: two case reports and literature review. *Orphanet journal of rare diseases.* 2008;3:19.
- 24. Wetterau JR, Aggerbeck LP, Bouma ME, et al. Absence of microsomal triglyceride transfer protein in individuals with abetalipoproteinemia. *Science*. 1992;258(5084):999-1001.
- 25. Gündüz M, Özaydın E, Atar MB, et al. Microsomal triglyceride transfer protein gene mutations in Turkish children: A novel mutation and clinical follow up. *Indian journal of gastroenterology : official journal of the Indian Society of Gastroenterology*. 2016;35(3):236-241.
- 26. Fairfield CJ, Drake TM, Pius R, et al. Genome-Wide Association Study of NAFLD Using Electronic Health Records. *Hepatology communications.* 2022;6(2):297-308.
- 27. Vujkovic M, Ramdas S, Lorenz KM, et al. A multiancestry genome-wide association study of unexplained chronic ALT elevation as a proxy for nonalcoholic fatty liver disease with histological and radiological validation. *Nature genetics.* 2022;54(6):761-771.

- 28. Khatun I, Walsh MT, Hussain MM. Loss of both phospholipid and triglyceride transfer activities of microsomal triglyceride transfer protein in abetalipoproteinemia. *Journal of lipid research.* 2013;54(6):1541-1549.
- 29. Li L, Wang SJ, Shi K, Chen D, Jia H, Zhu J. Correlation between MTP -493G>T polymorphism and non-alcoholic fatty liver disease risk: a meta-analysis. *Genet Mol Res.* 2014;13(4):10150-10161.
- 30. Musso G, Gambino R, Cassader M. Lipoprotein metabolism mediates the association of MTP polymorphism with beta-cell dysfunction in healthy subjects and in nondiabetic normolipidemic patients with nonalcoholic steatohepatitis. *J Nutr Biochem.* 2010;21(9):834-840.
- 31. Di Filippo M, Varret M, Boehm V, et al. Postprandial lipid absorption in seven heterozygous carriers of deleterious variants of MTTP in two abetalipoproteinemic families. *Journal of clinical lipidology*. 2019;13(1):201-212.
- 32. Di Filippo M, Crehalet H, Samson-Bouma ME, et al. Molecular and functional analysis of two new MTTP gene mutations in an atypical case of abetalipoproteinemia. *Journal of lipid research.* 2012;53(3):548-555.
- 33. Welty FK. Hypobetalipoproteinemia and abetalipoproteinemia: liver disease and cardiovascular disease. *Current opinion in lipidology*. 2020;31(2):49-55.
- 34. Pelusi S, Baselli G, Pietrelli A, et al. Rare Pathogenic Variants Predispose to Hepatocellular Carcinoma in Nonalcoholic Fatty Liver Disease. *Scientific reports.* 2019;9(1):3682.
- 35. Taylor R. Pathogenesis of type 2 diabetes: tracing the reverse route from cure to cause. *Diabetologia.* 2008;51(10):1781-1789.
- 36. Luukkonen PK, Zhou Y, Nidhina Haridas PA, et al. Impaired hepatic lipid synthesis from polyunsaturated fatty acids in TM6SF2 E167K variant carriers with NAFLD. *Journal of hepatology.* 2017;67(1):128-136.
- 37. Bianco C, Tavaglione F, Romeo S, Valenti L. Genetic risk scores and personalization of care in fatty liver disease. *Current opinion in pharmacology*. 2021;61:6-11.
- 38. Nakajima K, Nakano T, Tokita Y, et al. Postprandial lipoprotein metabolism: VLDL vs chylomicrons. *Clinica chimica acta; international journal of clinical chemistry*. 2011;412(15-16):1306-1318.
- 39. Ajmera V, Park CC, Caussy C, et al. Magnetic Resonance Imaging Proton Density Fat Fraction Associates With Progression of Fibrosis in Patients With Nonalcoholic Fatty Liver Disease. *Gastroenterology*. 2018;155(2):307-310.e302.
- 40. Schwabe RF, Tabas I, Pajvani UB. Mechanisms of Fibrosis Development in Nonalcoholic Steatohepatitis. *Gastroenterology*. 2020;158(7):1913-1928.
- 41. Friedman SL, Neuschwander-Tetri BA, Rinella M, Sanyal AJ. Mechanisms of NAFLD development and therapeutic strategies. *Nature medicine*. 2018;24(7):908-922.
- 42. Liu Y, Conlon DM, Bi X, et al. Lack of MTTP Activity in Pluripotent Stem Cell-Derived Hepatocytes and Cardiomyocytes Abolishes apoB Secretion and Increases Cell Stress. *Cell reports.* 2017;19(7):1456-1466.
- 43. Reich D, Thangaraj K, Patterson N, Price AL, Singh L. Reconstructing Indian population history. *Nature.* 2009;461(7263):489-494.

Tables

Table 1. Clinical characteristics of family members.

Person (Fig 1A)	<i>MTTP</i> p.Ile564Thr	Sex	Liver Disease Diagnosis (method)	Age Diagnosed	Type 2 diabetes	BMI	Hypertension	Liver biochemistry & lipid blood analyses at diagnosis	Other subsequent investigations, treatments and comorbidities	PNPLA3 p.1148M	TM6SF2 p.E167K	<i>MTTP</i> p.1128T	<i>MTTP</i> р.Q297Н
В	IT	4	Cirrhosis & ascites (USS)	88	X	<30	✓	ALT=26, AP=120, Chol=3.5, TG=1.29,	Cardiovascular-related death aged 90.	MM	EK	TT	HH
С	ΤT	0+	HCC & cirrhosis (CT/Biopsy)	57	✓	<30	✓	ALT=38, AP=129, Normal lipids,	Liver resection ablation, right hemihepatectomy, HCC recurred. Liver-related death.	IM	EE	TT	ΗH
D	TT	0	Cirrhosis (MRI)	60	X	<30	✓	Bilirubin=312µmol/L, Normal lipids,	Liver screen=normal. Died of gallbladder sepsis aged 72.	IM	EE	TT	HH
E	IT	6	FL (USS)	57	~	30	✓	ALT=30, AP=75, Chol=2.5, TG=0.75, HDL=1.2, LDL=1.0	Stable for 18y: TE=5.8kPa; CAP=341; ApoB=0.94	IM	EE	nd	ΗН
F	TT	50	Cirrhosis (USS)	56	X	21	~	ALT=23, AP=125, Chol=2.4, TG=0.58, HDL=1.7, LDL=0.4,	No retinitis pigmentosa. Vagotomy and pyloroplasty; Mild Bone marrow failure;	IM	EK	TT	нн
G	T*1	8	HCC & cirrhosis (Biopsy)	57	X	<30	X	Normal lipids	Multifocal HCC: Chemotherapy and Hepatectomy. Died aged 61.	nd	nd	nd	nd
H	ΤT	8	HCC, cirrhosis, Portal hypertension	51	х	<30	~	Chol=2.2, TG=0.2, HDL=1.6, LDL=0.5,	Liver transplant x2. Died due to cardiac complications aged 63.	IM	EK	TT	nd
Ι	Π	4	FL (USS)	50	X	<30	X		Diet/Lifestyle changed: recovered/stable >5y:TE=4.0kPa; CAP=223 ApoB=1.02	IM	EE	TT	ΗН
J	TT	4	Cirrhosis (MRI) HCC	50 54	х	25	~	ALT=52, AP=87, ApoB=0.32, Normal lipids;	Liver transplant age 56: recovered/stable >5y:TE=4.6kPa CAP=214; BMI=20.3; Duodenal Biopsy: no evidence of abetalipoproteinemia	IM	EE	TT	нн
K	IT	4	NASH (USS) (TE:CAP=38 0)	48	х	24	~	ALT=79, AP=108; Normal lipids; ApoB=1.0;	Diet/Lifestyle changed: recovered/stable 10y:TE=6.2kPa CAP=298. Duodenal Bx: no evidence of abetalipoproteinemia	IM	EK	nd	ΗH
L	TT	4	Cirrhosis (USS)	59	х	18	~	Elevated ALT & AST Normal lipids	Stable for 10y: TE=10.3kPa: CAP=276, ALT=104, AP=87, AST=73, Cardiac arrhythmia, osteoporosis, Vitamin D deficiency	IM	EE	TT	ΗH
Μ	Π	0	Healthy	Х	х	26	х	Normal lipids	TE=4.6kPa; CAP=253; Vitamin D deficiency	ΙI	EK	IT	QH
N	IT	6	NASH (USS)	40	X	28	X	TE=3.6kPa CAP=373; lipids and ALT elevated	Lifestyle & diet modified – stable 7y	IM	EK	nd	QH
0	IT	8	FL (USS)	38	X	27	X	lipids elevated	Lifestyle & diet modified – recovered; Stable for 19y:TE=5.8kPa; CAP=341)	ΙI	EK	IT	QH
Q	ΙT	8	FL (USS)	28	X	19	х	Normal lipids, Duodenal Biopsy: evidence of duodenal lipid	Improved after 10y:TE=6.3kPa CAP=199.	MM	EE	IT	HH

For MTTP p.E98D all tested were EE; for MTTP p.N166S all tested were NN. 1likely T carrier based on pedigree. USS: abdominal ultrasonographic steatosis score; Bx: biopsy; FL: fatty liver; HCC: hepatocellular carcinoma; ALT: alanine transaminase U/L; AP: alkaline phosphatase U/L; Chol: cholesterol mmol/L; TG: triglycerides mmol/L; HDL: high-density lipoprotein mmol/L; LDL: low-density lipoprotein mmol/L; ApoB: apolipoprotein B g/L; TE: transient elastography; CAP: Controlled Attenuation Parameter dB/m; nd, not determined.

Table 2. Clinical characteristics of individuals investigated in generation IV.

ype	nale		ng	stes		nc	
MTTP genoty p.I564T	Sex male/fen	Liver Disease Diagnosis (method)	Age at testi	Type 2 diabe	BMI	Hypertensid	Liver biochemistry & lipid blood analyses at diagnosis
IT	0	NASH (USS)	32	х	20	х	ALT=53, AP=68, Chol=7, TG=3.14, VitD<15; TE=7.4kPa CAP=352
IT	03	FL (USS)	37	х	18	х	Normal LFTs (ALT=14), Normal Lipids; HbA1c=9.5;
IT	5	FL (USS)	40	х	28	X	ALT=74, Chol=6, TG=2.12; TE=3.6kPa CAP=373
IT	0 +	healthy (USS)	29	х	20	X	Normal LFTs (ALT=13), Normal Lipids ApoB=0.95, TE=3.5kPa CAP=122
ΙΤ	50	healthy (TE)	32	х	27	x	TG=2.57, Chol=3.7, ApoB=1.04. TE=5.3 kPa
ΙΤ	50	healthy (USS)	31	х	18	х	Normal LFTs (ALT=19), Normal lipids,
ΙΤ	Ŷ	healthy (USS)	37	х	19	х	Normal LFTs (ALT=10), Chol=4.2, TG=1.77; ApoB=0.96; TE=4.2kPa CAP=291
ΙΤ	4	healthy (USS)	39	х	21	х	Normal LFTs (ALT=23), Normal Lipids;
I T ¹	Ŷ	healthy (USS)	41	х	20	х	Normal LFTs (ALT=27), Chol=4.6, TG=2.15; ApoB=1.02, TE=3.1kPa CAP=231
I T ¹	0	healthy (USS)	44	X	<30	X	Normal LFTs (ALT=27), Chol=3.5, TG=2.4; TE=5.5kPa CAP=281
IT	Q	healthy (USS)	49	Х	<30	Х	Normal LFTs (ALT=14), Chol=4.88, TG=0.98, ApoB=1.09, TE=5.1kPa CAP=245
IT	0+	FL (USS) healthy (USS)	35 38	X	21	X	Normal LFTs (ALT=12), Chol=6, TG=1.5; ApoB=1.31, TE=4.2kPa CAP=325. Recovered after 3y: TE=6.3kPa CAP=278
ΙΤ	0	FL (USS)	38	х	27	х	Normal LFTs (ALT=39), Chol=2.9, TG=0.98;
T T ¹	2	nealthy (USS)	42	v	21	v	Recovered after 4y: TE=5.8kPa CAP=243 Normal L FTs (ALT=15) Normal Lipids
	0	healthy (USS)	14	Х	<i>∠</i> 1	Å	Recovered after 5y: TE=6.9kPa CAP=216
I I ¹	Ŷ	healthy (USS)	17	x	20	x	Normal LFTs (ALT=15), Normal Lipids TE=6.4kPa CAP=216
I I ¹	Ŷ	healthy (USS)	20	X	<30	х	Normal LFTs (ALT=9), Normal Lipids

¹deduced from pedigree. USS: abdominal ultrasonographic steatosis score; NASH: nonalcoholic steatohepatitis; FL: fatty liver; ALT: alanine transaminase U/L; AP: alkaline phosphatase U/L; Chol: cholesterol mmol/L; TG: triglycerides mmol/L; LFTs: liver function tests; ApoB: apolipoprotein B g/L (research laboratory data); TE: transient elastography; CAP: Controlled Attenuation Parameter dB/m.

Figure Legends

Fig. 1. Identification of a pathogenic variant in large family with non-alcoholic fatty liver disease.

(A) Pedigree. Clinical features are described in Table 1. Diagnosis is indicated by shading: black, hepatocellular carcinoma (HCC); dark grey, cirrhosis; light grey, non-alcoholic steatohepatitis (NASH); Dashed lines indicate no investigations; * Exome sequenced. Blue letters indicates residue 564 in MTP. (B) Variant rs745447480 sequencing. (C) local environment of I564 on MTP–PDI interface (hydrophobic pocket (white), polar (green) and charged (red/blue) residues); (D) 564T variant and common nonsynonymous variants (side-chains: C=cyan; O=red; N=blue) in model derived from PDB ID:617S⁸, a heterodimer of Protein Disulfide Isomerase (PDI), blue, and MTTP gene product (MTP), grey.

Fig. 2. Meal-response study to investigate metabolism in family members and matched controls.

(A) Study Design. (B) ApoB levels. Participants are grouped according to age and gender matching to family members F, J, K, Q and M (Supplementary Table 1). Participants are grouped according to age and gender-matched (white bars are healthy volunteers (HV); grey bars are NAFLD patients). MTP residue 564 is indicated and PNPLA3 or TM6SF2 in parentheses if homozygous for variant rs738409 or rs58542926. (C) Serum ApoB-100. HV, healthy volunteers; NAFLD, non-alcoholic fatty liver disease.

Fig. 3. Triglyceride levels in study participants.

Total serum triglycerides: (A), Participant F and matched controls and (B), J and matched controls. VLDL-triglyceride: (C), Participant F and matched controls and (D), J and matched controls. Chylomicron-triglyceride: (E) Participant F and matched controls and (F), J and matched controls. *MTTP* genotypes are shown for family members. *PNPLA3* p.I148M and *TM6SF2* p.E167K genotype are indicated in parentheses.

Fig. 4. Characterisation of MTP-564T homozygote variant, *MTTP*^(VAR/VAR) and wild-type hepatocyte-like cells (HLCs).

(A) Light microscopy image of terminally differentiated hiPSC-derived HLCs. (B) Oil-Red-O staining of HLCs (light microscopy). (C) Nile Red staining of lipids ±DAPI staining (fluorescence microscopy). (D) MTP protein expression immunocytochemistry ±DAPI staining. Quantification of Nile Red fluorescence (E), and MTP staining (F) in HLCs. (G) Expression of MTP determined by Q-PCR. (H) ApolipoproteinB-100 secretion by HLCs (ELISA). (I) Basal and maximal mitochondrial respiratory rates in *MTTP*^(WT/WT) (•) and *MTTP*^(VAR/VAR) (□) HLCs. Quantification of cellular superoxide (J) and reactive oxygen species (K) from fluorescence microscopy. Mean±standard error. P<0.05 is significant (T-test, paired, 2-tailed).

Fig. 5. Phenotypic characterisation of *MTTP*^(VAR/VAR) hiPSC-derived HLCs.

(A) Expression of inflammation related and ER-stress related genes. (B) Expression of *NFκB*-associated intracellular signalling proteins in *MTTP*^(VAR/VAR) HLCs, relative to expression in *MTTP*^(WT/WT). (C) Phosphorylated proteins in *MTTP*^(VAR/VAR) HLCs, normalised to *MTTP*^(WT/WT). (D) Secreted proteins from *MTTP*^(VAR/VAR) HLCs, normalised to *MTTP*^(WT/WT).

Fig. 6. Restoration of activities by gene editing of *MTTP*^{VAR/VAR} 564-TT to 564-II.

(A) Light microscopy showing terminally differentiated hiPSC-derived hepatocyte-like cells (HLCs) from *MTTP*^(VAR/VAR) and gene-edited derivative *MTTP*^(WT*/WT*). (B) MTP enzyme activity in HLCs. Mean±standard error; significance level P<0.05 (T-test).





¹milk and cornflakes.

²80-175min between breakfast start and pre-lunch sample.

³from participants J and 1 used to derive cell lines.

⁴see supplemental methods













MTTP^(var/var) MT

Α







Identification and characterisation of a rare MTTP variant underlying hereditary non-alcoholic fatty liver disease.

Hepatocyte-like cell model harbouring rs745447480 has reduced MTP lipoprotein secretion activity and develops cytoplasmic triglyceride accumulation affecting cell pathways, providing insights into disease mechanisms.

